

Author's response to reviews

Title: Quality of the Parent-Child Interaction in Young Children with Type 1 Diabetes Mellitus: Study Protocol

Authors:

Anke M Nieuwesteeg (a.m.nieuwesteeg@uvt.nl)
Frans Pouwer (f.pouwer@uvt.nl)
Hedwig JA van Bakel (h.j.a.vanbakel@uvt.nl)
Wilco HM Emons (w.h.m.emons@uvt.nl)
Henk-Jan Aanstoot (h.j.aanstoot@diabeter.nl)
Roelof Odink (roelof-odink@catharina-ziekenhuis.nl)
Esther E Hartman (e.e.hartman@uvt.nl)

Version: 5 **Date:** 17 March 2011

Author's response to reviews:

References 41 is now cited within the main text.

We have included the city in all of the affiliation details.

Methods: We have renamed this section 'Methods/Design'.

Additional file: We have removed the heading 'Additional files' and 'Additional file 1' from the figure legend.